Author’s response to reviews

Title: Suppression of low-density lipoprotein oxidation, vascular smooth muscle cell proliferation and migration by a herbal extract of Radix Astragali, Radix Codonopsis and Cortex Lycii

Authors:

Judy Y Chan (judychanyw@hotmail.com)
Johnny C Koon (johnnykoon@gmail.com)
Ping-Chung Leung (pingcleung@cuhk.edu.hk)
Chun-Tao Che (chect@cuhk.edu.hk)
Kwok-Pui Fung (kpfung@cuhk.edu.hk)

Version: 4 Date: 4 March 2011

Author’s response to reviews: see over
4 March 2011

BMC Complementary and Alternative Medicine

Dear Editor,

1952403058440251 - Suppression of low-density lipoprotein oxidation, vascular smooth muscle cell proliferation and migration by a herbal extract of Radix Astragali, Radix Codonopsis and Cortex Lycii

Thank you very much for your reply dated 11 February, 2011 in which you recommend us to revised Figure 5 and do the proofreading. As shown in the revised manuscript, we have replaced Figure 5 by the raw blot and used boxes to show the involved data. Moreover, we have edited the whole manuscript with grammatical errors and typos corrected.

We hope that you will consider accepting this manuscript for publication in BMC Complementary and Alternative Medicine.

Thank you for your attention.

Yours Sincerely,

KP Fung (Professor)
Corresponding author
School of Biomedical Sciences
The Chinese University of Hong Kong
Shatin, N.T., Hong Kong
Tel: 852-26096873
Fax: 85226037732
Email: kpfung@cuhk.edu.hk